Author's response to reviews

Title: Does surgery followed by physiotherapy improve short and long term outcome for patients with atraumatic shoulder instability compared with physiotherapy alone? Protocol for a randomized controlled clinical trial

Authors:

Anju Jaggi (Anju.Jaggi@rnoh.nhs.uk)
Susan Alexander (Susan.Alexander@rnoh.nhs.uk)
Robert Herbert (r.herbert@neura.edu.au)
Lennard Funk (lenfunk@shoulderdoc.co.uk)
Karen A Ginn (karen.ginn@sydney.edu.au)

Version: 5 Date: 7 October 2014

Author's response to reviews:

Dear Mr Aldea,

Please find attached our responses (in blue) to the reports from Reviewers regarding our revised manuscript “Does surgery followed by physiotherapy improve short and long term outcome for patients with atraumatic shoulder instability compared with physiotherapy alone? Protocol for a randomized controlled clinical trial.”

We look forward to hearing from you,

Karen Ginn
(On behalf of Ms Jaggi, Dr Alexander, Prof Herbert & Prof Funk)